Author's response to reviews

Title: Plasma branched-chain and aromatic amino acid concentration after ingestion of an urban or rural diet in rural Mexican women.

Authors:

- Adriana M. López (adrimar24@gmail.com)
- Lilia G. Noriega (lgnoriegal@gmail.com)
- Margarita Diaz (maggiediazm@yahoo.com)
- Nimbe Torres (nimbester@gmail.com)
- Armando R. Tovar (tovar.ar@gmail.com)

Version: 5  Date: 13 January 2015

Author's response to reviews:

Please find enclosed a new version of our manuscript, where we have modified the text according to the reviewer concerns, as well as a detailed point-by-point answer letter addressing all issues raised by the reviewer.